RESEARCH ARTICLE

Open Access



Eating behaviour disorders among adolescents in a middle school in Dongfanghong, China

Tingting Feng¹ and Dawit Shawel Abebe^{2*}

Abstract

Background: Disordered eating behaviours (DEB) are significant public health challenges among adolescents. DEB are prevalent among Chinese young people and replication epidemiological studies in DEB are needed due to ongoing rapid modernization and socio-economic change. In addition, there has been no prevention or intervention programs for DEB in most of rural areas in China and research in DEB in rural China is limited. More research in DEB in rural China is needed for increased awareness of prevention strategies. The objectives of the study are to examine the prevalence and associated factors of DEB among rural Chinese adolescents.

Methods: Four hundred and sixty-six students aged 11–18 years old from a middle school in rural China were invited to complete a self-reported questionnaire that included measures on DEB and psychosocial factors. The SCOFF questionnaire was applied to measure DEB (i.e., a score of > = 2 indicates a likely case of DEB). Logistic regression models were applied for data analyses. A p-value <0.05 was regarded as statistically significant.

Results: There were 389 adolescents (11–18 years) included in this study with the prevalence of DEB being 28.8%. No significant differences were found between male adolescents (30.5%) and female adolescents (27.1%). Independently, perceived overweight (OR = 2.80, 95% CI = 1.05–7.48), negative affect (OR = 1.07, 95% CI = 1.03–1.12), body dissatisfaction (OR = 0.96, 95% CI = 0.93–0.99), and watching TV (OR = 2.40, 95% CI = 1.11–5.18) were significantly (p < 0.05) associated with DEB.

Conclusion: This study indicates a high prevalence rate of DEB among rural adolescents and associated factors of DEB from a school-based sample. Such findings imply that local public health systems should pay more attention to DEB and design prevention and intervention strategies for DEB.

Keywords: Disordered eating behaviours, Associated factors, Adolescents, China

Plain English summary

Disordered Eating Behaviours (DEB) start to appear during adolescence and bring serious consequences. Previous studies showed DEB is becoming common in China, and even more prevalent than that is in Western countries. However, limited research has been conducted among adolescents in rural China. The current study aimed to establish the prevalence of DEB and associated psychosocial factors among rural adolescents. Results from the current research showed a high prevalence rate

of DEB among rural adolescents and associated psychosocial factors of DEB.

Background

Disordered Eating Behaviours (DEB) are abnormal eating attitudes and behaviours which have much in common with diagnosed eating disorders [1]. These can have serious implications which relate to depression, substance abuse, suicidal behaviours, social isolation and which may develop into an eating disorder [2, 3].

DEB has long been regarded as strongly related to Western cultures but according to a review in the most recent published epidemiologic data relevant to eating disorders in Asia and the Pacific, abnormal eating

²Norwegian Social Research (NOVA), Oslo and Akershus University College of Applied Sciences, Oslo, Norway





^{*} Correspondence: dawit.s.abebe@nova.hioa.no

behaviours found to be highly prevalent in China [4]. Traditional Chinese culture considers plumpness as beautiful and slimness as reflecting poverty. However, China has been dramatically influenced by Western cultures for the past 39 years due to the 'open-door policy'. The thin-ideal has come to overshadow standards for beauty which contributes to body dissatisfaction and DEB among females [5]. Accordingly, a number of Chinese studies report a high prevalence of DEB [6-13]. Specifically, female adolescents living in Hong Kong have been found to have the highest prevalence of DEB (10.8%), followed by those from Shenzhen (5.2%) and from Hunan (2.5%) [6]. Two school-based studies in Taiwan revealed DEB in 17.1% and 10.4% of high school female adolescents [7, 8]. Further, a school-based study in Hong Kong revealed that DEB were present in 3.9% of adolescent males and 6.5% of adolescent females [9]. Three school-based studies in Mainland China have also reported that the prevalence of DEB varied from 2.3% to 17% among middle school and college students [10-12]. Western studies have revealed DEB in 29.4% of female adolescents, specifically 14.9% of male adolescents in Germany, and 56% of female adolescents and 28% of male adolescents in US [14, 15]. However, few studies in DEB have included male adolescents in China [9, 13]. The variations in measuring instrument, sample characteristics, methodological approaches and study areas, however, may limit the comparability of prevalence across studies. The updated epidemiologic review in 2016 suggested that fat concern and weight control behaviours are prevalent among Chinese young people and replication epidemiological studies in eating disturbances are needed due to ongoing rapid modernization and socioeconomic change [4].

Psychosocial risk factors are strongly associated with a high prevalence of DEB among adolescents [13, 16]. For instance, longitudinal studies conducted in Southern China found body dissatisfaction, negative affect, perceived sociocultural pressures, and high socioeconomic status (SES) all have significant associations with DEB. [13, 17].

A comparative study among female adolescents in three different socioeconomically communities of China documented that those in rural areas had fewer concerns about body fat and demonstrated a lower prevalence rate of DEB than those in developed cities [6]. Socioeconomic development in China is growing very fast nowadays and the rapid ongoing socioeconomic transition has influenced rural China to a large extent. Rural populations now have more opportunities to access the mass media, which may relate to a higher risk of DEB. T Jackson and H Chen have reported that Chinese middle adolescent girls reported higher levels of appearance experiences related to mass media and DEB [18], and pressure from mass media related to DEB have been observed in Chinese adolescent

boys [19]. There are some studies examining the epidemiology of eating disorders in rural China [17, 20, 21], but little research has focused on epidemiology of DEB in rural China. In more than 20 years, there has been only one study which has focused on DEB among Chinese female adolescents in a rural area [6]. Further, given that little attention has been paid to DEB in both genders in rural China, it is necessary to re-investigate the situation under current influences. The present study, therefore, aimed to examine prevalence and associated psychosocial factors of DEB among male adolescents and female adolescents at a Chinese middle school in a rural area. This may contribute to the updated information in the epidemiology of DEB in China and also respond to the updated review's request for replication studies, that is this review suggested further replication epidemiological studies in eating disturbances due to ongoing rapid modernization and socio-economic change [4].

Methods

Participants and study procedure

All students from a middle school – a secondary high school – in Dongfanghong town (a less economically developed rural area), Heilongjiang province were invited to participate in the study. In 2015, the per capita annual income in our sample area was 30,977 RMB (or \$4509). The school that was studied was the only middle school in this area. Four hundred and sixty-six students aged 11–18 years old were invited to complete an anonymous, self-reported questionnaire covering a wide range of topics including DEB and related psychosocial behaviours on a voluntary basis.

Before starting the research, schoolteachers were trained by researchers and schoolteachers explained the study to students and were able to answer students' questions. During the study, the confidentiality and the anonymity of participants were assured. With the assistance of schoolteachers, the questionnaires were administered in September 2015 - at the beginning of autumn semester. The class teachers decided the time when questionnaires were conducted. They usually selected self-learning class (students studied by themselves) and the time of self-learning class was different among classrooms. The students used around 1 h -2 h to finish the questionnaires. The class teachers collected the questionnaires following completion and returned them to the researcher. Of 466 questionnaires that were issued, 389 valid questionnaires were returned (a response rate of 83.5%).

Outcome variables

SCOFF: The SCOFF questionnaire is an effective tool that can screen DEB quickly [22]. It includes five items addressing the main features of DEB [22]. See the

Additional file 1: Table S1 for the descriptive statistics of each item.

One point for every "yes"; a score of > = 2 indicates a likely case of DEB. The scale has been validated and has demonstrated effective specificity in identifying DEB [23]. The Chinese version of the SCOFF questionnaire has been proved to be effective for detecting potential DEB [24].

Independent variables

Body Mass Index (BMI) was computed from self-reported measures of height and weight. Self-reported BMI has been demonstrated to be a reliable measure of BMI [25]. The classification of BMI was made according to BMI cut-offs by the Working Group on Obesity in China (WGOC) [26]: $\leq 18.49 \text{ kg/m}^2$ (underweight), $18.50-23.99 \text{ kg/m}^2$ (normal weight), 24-27.99 (overweight), ≥ 28 (obesity). We combined the overweight group and the obesity group into one category due to the small number size in the obesity group.

Participants were asked to describe their weight perceptions. Responses were ranged into three groups: underweight, average /normal, and overweight.

Compared weight perceptions with actual BMI, weight misconception was ranged into four groups: misconception of underweight, overweight, normal weight and correct weight conception. See the Additional file 2: Table S2 for the association between BMI and perceived weight categories.

Body dissatisfaction was examined using a Body Part Satisfaction Scale with each response from 1 = extremely dissatisfied to 6 = extremely satisfied [27]. The scale consists of eight items rating the individual's level of satisfaction with the following body areas: face, upper torso, middle torso, lower torso, muscle tone, weight, height and overall appearance. A high score indicates a higher level of body satisfaction. The scale has shown reliable psychometrics [27]. The Chinese version of this scale has been tested and has strong reliability [18]. The internal consistency of the scale was 0.90.

Depressive symptoms were measured by the Positive and Negative Affect Schedule [28]. This schedule is a 20-item test with 10 items for negative affect and positive affect, respectively. Participants were asked to describe the extent to which they had varying feelings and emotions in the past month from 1 = very slightly or not at all to 5 = extremely. The scale has shown good validity and acceptable psychometric properties [28]. The original scale has been tested on Chinese samples [19].

The Perceived Sociocultural Pressure Scale was used to assess pressures to be thin from family, friends, dating partners and media [18]. The questionnaire has 10 items that investigate perceived sociocultural pressures to be thin. An example of an item is: "I've noticed a strong message from my family to have a thin body". Three items are for family and friends, respectively. Two items

are for dating partners and media, respectively. Responses ranged from 1 = not at all to 5 = extremely. Higher scores indicate more perceived pressure. Internal consistency ($\alpha = 0.83$), test-retest reliability (r = 0.93) and predictive validity of this scale have been documented [29]. The scale has been tested and used in Chinese samples [19].

Media use was assessed by a media exposure questionnaire. Participants were asked: whether they watch television and DVDs, whether they use the Internet, whether they use chat rooms and play computer games, as well as how long they spent on media. The questionnaire was developed from a Chinese study [5].

Parental socioeconomic status (SES) was determined on the adolescents' report of parents' occupation. Occupations were categorized according to the classification by the Chinese National Bureau of Statistics [30] with nine items to be selected. Four groups were created based on the salary of occupations: professional, service, labourers, and unemployed. The variables of mother's job and father's job were merged into one variable (parents' occupation), recording the higher of the two values.

Age and sex were also recorded.

Statistical analysis

Chi-square analysis was used to compare differences in prevalence of DEB and for comparing perceived weight categories to BMI categories. To examine associated factors of DEB, first, a univariate logistic regression was employed to examine the association between each of the independent variable and DEB, and then multiple logistic analysis was performed. A forward stepwise multiple logistic regression was performed to determine factors significantly associated with DEB, including independent variables based on univariate logistic regression results (cut-off point $p \le 0.25$). Age and gender were controlled. The final model was built based on this forward stepwise multiple logistic regression ($p \leq 0.05$). The multi-collinearity among associated factors was examined with collin command in Stata (mean variance inflation factor (VIF) = 1.43, correlation matrix = 0.0528). Secondly, Model fit to the data (validity and reliability) was assessed by the likelihood ratio χ^2 test, Hosmer-Lemeshowgoodness-of-fit test and receiver operating characteristics (ROC) curve. $P \le 0.05$ was regarded as statistically significant and odds ratio (OR) with 95% confidence intervals (95% CI) were calculated. Data analyses were conducted with the Statistical Package for the Social Sciences (SPSS) 22.0 and StataSE 14.

Ethical clearance

The study was approved by the Regional Committee for Medical Research Ethics in Norway, and the Committee for the middle school in Dongfanghong, China.

Results

Descriptive summary of the study participants

Table 1 presents the descriptive summary of study participants (N = 389).

Prevalence of disordered eating behaviours

Table 2 presents the prevalence of DEB according to the participants' general information by using chi-square analysis. The prevalence of DEB among adolescents was 28.8%. The prevalence of DEB among male adolescents and female adolescents was 27.1% and 30.5%, respectively; no significant differences were found based on gender. No significant differences of prevalence of DEB were found across different age groups. Significant differences in the prevalence of DEB were found in different BMI groups, χ^2 (2) = 14.896, p = .001. Overweight adolescents (42.9%) had a significantly higher prevalence of DEB as compared to underweight adolescents (18.6%). Significant differences in the prevalence of DEB were also found in different perceived weight groups, χ^2 (2) = 51.524, p < .001. Adolescents in the perceived overweight group (54.4%) had a significantly higher prevalence of DEB than those among the perceived underweight group (18.0%). Significant differences in the prevalence of DEB were found in different weight misconception groups, χ^2 (2) = 43.399, p < .001. Adolescents with overweight misconception (59.4%) had a significantly higher prevalence of DEB than those without overweight misconception. There were no significant differences in the prevalence of DEB among students across different SES groups. Associations between misconception of over-/under-weight and DEB by gender is shown in the Additional file 3: Table S3.

Psychosocial factors independently predicting disordered eating behaviours

Table 3 presents results from univariate and multiple logistic regression analyses reporting psychosocial associated factors of DEB. The final multiple regression model included perceived weight, body dissatisfaction, negative affect and watching TV. The risk of having DEB among the perceived overweight group was 2.8 times higher than among those in the perceived underweight group. Students with higher levels of body dissatisfaction were significantly more likely to develop DEB compared to those with lower levels of dissatisfaction. Students with a higher negative affect were significantly more likely to develop DEB compared with those with a lower negative affect. Students watching TV were almost three times more likely to report DEB than students who did not watch TV. The corresponding coefficients and standard errors in the multiple logistic regression model are shown in the Additional file 4: Table S4.

The reliability and the validity of the multiple logistic regression model was assessed by employing postestimation statistics (likelihood ratio χ^2 test significant, $\chi^2 = 107.925$, degrees of freedom (df) = 9, p < 0.001; Hosmer-Lemeshow χ^2 test = 10.196, df = 8, p = 0.252). The Hosmer-Lemeshow test of the goodness-of-fit suggests the model is a good fit to the data (small Chisquared values with larger p-value closer to 1). Area under the ROC curve (AUC) was 0.825 with 95% confidence interval (0.781, 0.869), which is a prediction power for the model.

Discussion

The main findings of this school-based study reveal a high prevalence rate of DEB among male adolescents (27.1%) and female adolescents (30.5%). The study also found that perceived weight, negative affect, body dissatisfaction, and watching TV were significantly associated with DEB among adolescents.

The prevalence estimates of the current study are among a wide range of previous school-based studies conducted in China (2.26%-58.3%) [10, 31-33]. The available information on DEB in rural China has been reported by Sing Lee and Antoinette M. Lee [6]. In their study, 266 school female adolescents from Grade 11 from Rural Hunan were included with a prevalence rate of 2.5%. In the present study, prevalence was much higher. This may be attributed to different screening instruments for DEB. The study by Sing Lee and Antoinette M. Lee [6] used EAT-26 as the screening instrument, while we used SCOFF. The sensitivity and the specificity between the two instruments are different and SCOFF has been reported to have a tendency toward overinclusion [34]. It is further comparable to reports from Western countries; in spite of variation in measuring instruments, study methodologies and sample compositions: for instance 29.4% of female adolescents and 14.9% of male adolescents in Germany, and 56% of female adolescents and 28% of male adolescents in US [14, 15]. Given the harmful consequences related to DEB and the high prevalence of DEB, it is vital for public health officials to increase public awareness by introducing relevant and important information about DEB at individual, school and society level. In this way, health organisations can aim at intervening and preventing harmful eating behaviours. So far, there has been no such programs for DEB in the local region.

This study did not show significant gender differences in prevalence of DEB. Most Chinese and Western studies report that prevalence of DEB was significantly higher among female adolescents than among male adolescents [9, 14, 15]. The prevalence of DEB for males is possibly rising and the incidence of DEB for males may have been underrated due to cultural stigma [35]. It may also due to changing lifestyles: increased access to TV,

Table 1 Descriptive summary of the study participants (N = 389)

Variables	N	Percent	М	sd
Female	190	48.8		
SCOFF total score (five tiems)			1.0	1.1
0	172	44.2		
1	105	27.0		
2	67	17.2		
3	32	8.2		
4	12	3.1		
5	1	0.3		
Age(year)			15.1	1.7
Early Adolescence [11–13]	95	24.4		
Middle Adolescence [14-16]	198	50.9		
Late Adolescence [17, 18]	96	24.7		
BMI(kg/m ²)			20.5	4.7
Underweight	140	36.0		
Normal Weight	169	43.4		
Overweight	77	19.8		
Perceived Weight				
Underweight	61	15.7		
Normal Weight	214	55.0		
Overweight	114	29.3		
Weight Misconception				
Misconception of Underweight	18	4.6		
Misconception of Normal Weight	102	26.2		
Misconception of Overweight	57	14.7		
Correct Weight Conception	207	53.2		
Parents' Occupation				
Professional	115	29.6		
Service	72	18.5		
Labourers	48	12.3		
Unemployed	154	39.6		
With DEB	112	28.8		
Pressure				
Family			1.9	1.1
Friend			1.8	1.0
Dating			1.5	1.0
Media			1.6	1.1
Watch TV	323	83.0		
Watch DVD	74	19.0		
Surfing Online	353	90.7		
Chat Online	316	81.2		
Play Online	253	65.0		
Hours on Media average - 0-4.9 h	315	81		
Hours on Media Total			14.1	9.1
Body Satisfaction			31.1	8.4

Table 1 Descriptive summary of the study participants (N = 389) (Continued)

(COTITITACA)				
Variables	N	Percent	Μ	sd
Face			4.0	1.3
Upper torso			4.0	1.3
Middle torso			4.0	1.4
Lower torso			4.0	1.4
Muscle tone			3.7	1.4
Height			3.8	1.4
Weight			3.7	1.5
Overall appearance			4.1	1.2
Positive Affect			29.6	7.5
Negative Affect			21.2	7.2

N number, M mean, sd standard deviation

Pressure perceived sociocultural pressure to the

Pressure perceived sociocultural pressure to thin body and weight loss Hours on Media Total time on Internet working days + time on Internet weekends + time on TV working days + time on TV weekends Hours on Media average Hours on Media Total \div 4

social media, Internet and gaming. It is possible that female adolescents and male adolescents have a similar risk for developing DEB. More attention, therefore, should be also be paid to male adolescents who are at risk of DEB.

The present study found perceived weight as a significant associated factor of DEB. Results are consistent with previous observations on perceived weight status in Chinese adolescents [6] and Western samples [36, 37]. It has been suggested that one's perceived weight has a greater influence than one's actual weight when engaging in DEB [38]. In addition, a considerable number of adolescents in this study misperception with regard to weight status, which is consistent with previous Chinese research [39]. Furthermore, separate logistical regression in this study showed that students with the misconception of being overweight were more likely to report DEB. This is in line with previous research [38] and it indicates that one possible avenue for prevention programs may be a focus upon helping students develop accurate weight perception.

Along the same lines, body dissatisfaction was a significant associated factor for DEB. Similar results have been reported in both Chinese and Western research [13, 16]. The thin-ideal image contributes to perfectionism and body dissatisfaction. Male adolescents wish to become more muscular and achieve the male-ideal of V-shaped figure. On the other hand, female adolescents wish to become thinner and achieve the female-ideal of extremely thin. They put much emphasis upon having a perfect body shape, though they have a clinically healthy weight. Perfectionism and body dissatisfaction are both highly correlated with DEB [40]. Body dissatisfaction can increase dieting behaviours [41] and negative affect [16],

Table 2 Prevalence of disordered eating behaviours across participants' general information

	With DEB	Without DEB	Chi-square
	% (N)	% (N)	Statistic
Gender			0.545
Female	30.5 (58)	69.5 (132)	
Male	27.1 (54)	72.9 (145)	
Age			4.919
Early Adolescent [11–13]	24.2 (23)	75.8 (72)	
Middle Adolescent [14–16]	26.8 (53)	73.2 (145)	
Late Adolescent [17, 18]	37.5 (36)	62.5 (60)	
BMI			14.896**
Underweight	18.6 (26)	81.4 (114)	
Normal Weight	30.8 (52)	69.2 (117)	
Overweight	42.9 (33)	57.1 (44)	
Perceived Weight			51.524***
Underweight	18.0 (11)	82.0 (50)	
Normal Weight	18.2 (39)	81.8 (175)	
Overweight	54.4 (62)	45.6 (52)	
Weight Misconception			41.521***
Misconception of Underweight	22.2 (4)	77.8 (14)	
Misconception of Normal Weight	13.7 (14)	86.3 (88)	
Misconception of Overweight	61.4 (35)	38.6 (22)	
Correct Weight Conception	27.5 (57)	72.5 (150)	
Parents' Occupation			2.218
Professional	33.9 (39)	66.1 (76)	
Service	25.0 (18)	75.0 (54)	
Labourers	27.1 (13)	72.9 (35)	
Unemployed	27.3 (42)	72.7 (112)	

N number of participants p < 0.05, p < 0.01, p < 0.01

which can, in turn, lead to poor eating habits and increased eating disturbances.

Negative affect was significantly associated with DEB in this study. The result is in agreement with previous Chinese and Western research [13, 16]. One explanation is that negative urgency (i.e., the tendency to act rashly in response to negative affect) is significantly associated with disordered eating [42]. Another explanation can be derived from escape theory - people tend to escape negative emotions by abnormal eating (overeating) and they believe or have learned that overeating can relieve negative emotions [43, 44]. Moreover, negative affect and disordered eating may be interacted mutually, where the feeling of shame and guilty caused by disordered eating may result in negative self-perceptions among female adolescents [44].

In addition, this study found television access as a significant associated factor of DEB. A small number of adolescents did not have access to TV, possibly related to high academic expectations, family economic situations or personal preference. Media consumption can significantly predict female adolescents' DEB and male adolescents' interest in dieting [45]. Such an effect of media can also be found in other non-western studies. For example, research in Fiji [46], Pakistan [47], Japan [48], and Korea [49] have documented significant associations between media exposure and DEB. Becker [50] has also discussed how Western media introduced by TV has caused thinideal internalization among Fijian females. Garfinkel and Garner suggest: "The media have capitalized upon and promoted this image (of thinness) and through popular programming have portrayed the successful and beautiful protagonists as thin. Thinness has thus become associated with self-control and success." (p.145) [51]. Other media variables in this study did not show significant associations with DEB. It is possible that specific media exposure may contribute to thin body and eating disturbances instead of other general media variables.

Although BMI did not contribute to the multiple logistical regression model in this study, BMI was significantly positively associated with DEB from univariate logistical regression analysis. Previous research showed there may exist a U-shaped relation between DEB and BMI [52, 53]. That is, disordered eating behaviours were more problematic at the two extreme ends of BMI (underweight and obesity).

In addition, there is a non-significant trend of increasing prevalence of DEB with age in this study; late adolescents reported highest prevalence of DEB. This may be related to an increased social and emotional stress. Late adolescents, face an exceptional amount of pressure to succeed in college entrance examinations and achieve parental expectations [54] and experience peer competition, copious homework and reduced social support. These stressors may be associated with an increased risk for DEB [55].

This study adds to a growing literature on epidemiology of DEB among adolescents in China. It, therefore, provides current knowledge on the epidemiology of DEB in rural China, which helps to capture more attention to DEB in rural China. Since DEB may develop into more severe forms of DEB or eating disorders without appropriate intervention strategies, it is important to design early interventions against DEB. Secondly, this study included both male adolescents and female adolescents which has provided valuable insights into DEB in both genders. Thirdly, this study found risky behaviours associated with DEB among adolescents, which may contribute to better identification of high-risk groups and may be helpful for early prevention and intervention

Table 3 Multiple logistic regression model showing factors associated with disordered eating behaviours

	Univariate Logistic Regression Analysis per Independent Variable		Forward Multivariate Stepwise Logistic Regression Analysis	
	OR (95% CI)	Р	OR (95% CI)	Р
BMI				
Underweight	1		=	-
Normal Weight	1.95 (1.14–3.33)	.015**	=	-
Overweight	3.29 (1.77–6.11)	<.001****	-	-
Perceived Weight				
Underweight	1		1	
Normal Weight	1.01 (0.48–2.12)	.973	0.79 (0.34–1.86)	.589
Overweight	5.42 (2.56–11.47)	<.001****	2.80 (1.05-7.48)	.041**
Weight Misconception				
Correct Weight Conception	1		-	
Misconception of Underweight	0.75 (0.24–2.38)	.628	-	
Misconception of Normal Weight	0.42 (0.22-0.80)	.008***	-	-
Misconception of Overweight	4.19 (2.27–7.74)	<.001****	-	-
Pressure from family	2.17 (1.75–2.70)	<.001****	=	-
Pressure from friends	2.48 (1.93–3.18)	<.001****	=	-
Pressure from dating	1.86 (1.48–2.34)	<.001****	-	-
Pressure from media	1.90 (1.53–2.36)	<.001****	=	-
Watch TV				
No	1		1	
Yes	1.80 (0.94–3.46)	.076*	2.40 (1.11–5.18)	.026**
Watch DVD				
No	1		=	-
Yes	1.67 (0.98–2.85)	.058*	-	-
Chat Online				
No	1		-	-
Yes	2.35 (1.21–4.57)	.011**	_	_
Play Online				
No	1		_	-
Yes	1.34 (0.84–2.14)	.23*	_	_
Body Satisfaction	0.92 (0.89–0.95)	<.001****	0.96 (0.93-0.99)	.019**
Positive Affect	0.96 (0.93-0.99)	.005***	-	_
Negative Affect	1.10 (1.06–1.14)	<.001****	1.07 (1.03–1.12)	<.001****

OR odds ratio, CI confidence interval

Pressure perceived sociocultural pressure to thin body and weight loss $*p \le 0.25, **p < 0.05, ***p < 0.01, ****p < 0.001$

programs. For instance, adolescents with inaccurate weight perception and body dissatisfaction could be a targeted group for eating problems' prevention programs. A school-based program could be carried out by helping students reduce weight-related teasing, improve self-esteem and body satisfaction.

This study has some limitations, however. Firstly, the cross-sectional nature of this study could cause difficulty for examining cause-and-effect relationships between DEB and associated factors. Secondly, self-report measures in this study may lead to under- or overestimation of prevalence of DEB and associated factors. For example, self-reported BMI may overestimate measured BMI at the low end of the BMI scale and underestimate measured BMI at the high end [56]. Thirdly, unknown response bias may exist that can reduce the reliability and validity of the study. For example, students may respond untruthfully or give answers that they felt the researchers wanted to hear. Fourthly, the limitation of the SCOFF questionnaire needs to be considered. Since it has shown low sensitivity in a validation populationbased study [57]. Fifthly, information on students' reading level was not available. Since low reading levels may lead to misunderstanding of the questionnaires, the lack of data on reading levels may reduce the reliability of the study. Lastly, this study was school-based and thus access to participants was limited by those who were attending school. No accurate statistical information about school dropout rate was available. According to a literature review in 2015 [58], cumulative dropout rate in rural secondary schools was as high as 63%, while dropout rate in cities was less than 10%. There have been no statistics for students attending vocational and technical schools in China. This sample may be biased towards adolescents with higher education, therefore. The results would be more representative by conducting a population- or community-based study.

Conclusion

The present study has re-examined DEB among Chinese adolescents in a rural area. It suggests that DEB among rural Chinese adolescents are prevalent. Perceived weight, negative affect, body dissatisfaction and watching TV are significant associated factors for DEB among adolescents. This may provide valuable implications for prevention and intervention strategies in DEB. Furthermore, longitudinal population-based studies are recommended to produce a more comprehensive understanding of causes and risk factors of DEB among Chinese adolescents.

Additional files

Additional file 1: Table S1. Descriptive statistics of each item of SCOFF. (DOCX 11 kb)

Additional file 2: Table S2. Cross-tabulation of BMI and perceived weight categories among adolescents. (DOCX 11 kb)

Additional file 3: Table S3. Associations between misconception of over—/under-weight and DEB by gender. (DOCX 11 kb)

Additional file 4: Table S4. Multiple logistic regression model showing factors associated with disordered eating behaviours by coefficient and standard errors. (DOCX 13 kb)

Abbreviations

AUC: Area under the ROC curve; BMI: Body Mass Index; CI: Confidence intervals; DEB: Disordered eating behaviours; df: Degrees of freedom; M: Mean; N: Number; OR: Odds ratio; ROC curve: Receiver operating characteristics curve; sd: Standard deviation; SES: Socioeconomic status; SPSS: Statistical package for the social sciences; VIF: Variance inflation factor

Acknowledgements

The authors sincerely thank Li Chunying, Zhao Yanhua and Gong Chengzhi for help during research in the first middle school. The authors thank teachers' and students' in the first middle school for their cooperation.

Funding

This was a self-financed study.

Availability of data and materials

The data that support the findings of this study are available from the corresponding author upon reasonable request.

Authors' contributions

TF made substantial contributions to study design, acquisition of data, analysis and interpretation of data, and writing the manuscript. DSA made substantial contributions to study design, analysis and interpretation of data, revising the manuscript critically for important intellectual content and general supervision of the research. Both authors read and approved the final manuscript.

Ethics approval and consent to participate

The study was approved by the Regional Committee for Medical Research Ethics in Norway, and the Committee for the first middle school in Dongfanghong, China.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Author details

¹University of Oslo, Oslo, Norway. ²Norwegian Social Research (NOVA), Oslo and Akershus University College of Applied Sciences, Oslo, Norway.

Received: 2 June 2017 Accepted: 25 September 2017 Published online: 26 October 2017

References

- Association AP. Diagnostic and statistical manual of mental disorders (DSM-5°): American Psychiatric Pub; 2013.
- Shisslak CM, Crago M, Estes LS. The spectrum of eating disturbances. Int J Eat Disord. 1995;18(3):209–19.
- Haedt AA, Keel PK. Comparing definitions of purging disorder on point prevalence and associations with external validators. Int J Eat Disord. 2010; 43(5):433–9
- Thomas JJ, Lee S, Becker AE. Updates in the epidemiology of eating disorders in Asia and the Pacific. Curr Opin Psychiatry. 2016;29(6):354–62.
- Peat CM, Von Holle A, Watson H, Huang L, Thornton LM, Zhang B, et al. The association between internet and television access and disordered eating in a Chinese sample. Int J Eat Disord. 2015;48(6):663–9.
- Lee S, Lee AM. Disordered eating in three communities of China: A comparative study of female high school students in Hong Kong, Shenzhen, and rural Hunan. Int J Eat Disord. 2000;27(3):317–27.
- Chang Y-J, Lin W, Wong Y. Survey on eating disorder-related thoughts, behaviors, and their relationship with food intake and nutritional status in female high school students in Taiwan. J Am Coll Nutr. 2011;30(1):39–48.
- Tsai M-R, Chang Y-J, Lien P-J. Survey on eating disorders related thoughts, behaviors and dietary intake in female junior high school students in Taiwan. Asia Pac J Clin Nutr. 2011;20(2):196.
- Tam CK, Ng CF, Yu CM, Young BW. Disordered eating attitudes and behaviours among adolescents in Hong Kong: prevalence and correlates. J Paediatr Child Health. 2007;43(12):811–7.
- Lu X, Zhang N. A survey of eating behaviors among 703 female university and middle schoole students in Nanjing. Chin Sch Health. 2008;29:637–8.
- 11. Liao Y, Liu T, Cheng Y, Wang J, Deng Y, Hao W, et al. Changes in eating attitudes, eating disorders and body weight in Chinese medical university students. Int J Soc Psychiatry. 2013;59(6):578–85.
- Liang XM, Guo LT, Liu KZ. A cross-sectional investigation on eating disorders in 1486 female students from universities, senior high schools and junior high schools in Chengdu. Chin J Epidemiol. 2008;29(4):321–4.
- Jackson T, Chen H. Risk factors for disordered eating during early and middle adolescence: A two year longitudinal study of mainland Chinese boys and girls. J Abnorm Child Psychol. 2014;42(5):791–802.

- Wille D-PN, Hölling D-PH, Vloet TD, Ravens-Sieberer U. group Bs. Disordered eating behaviour and attitudes, associated psychopathology and healthrelated quality of life: results of the BELLA study. Eur Child Adolesc Psychiatry. 2008;17(1):82–91.
- Croll J, Neumark-Sztainer D, Story M, Ireland M. Prevalence and risk and protective factors related to disordered eating behaviors among adolescents: relationship to gender and ethnicity. J Adolesc Health. 2002;31(2):166–75.
- Stice E. Risk and maintenance factors for eating pathology: a meta-analytic review. Psychol Bull. 2002;128(5):825.
- Chen H, Jackson T. Prevalence and sociodemographic correlates of eating disorder endorsements among adolescents and young adults from China. Eur Eat Disord Rev. 2008;16(5):375–85.
- Jackson T, Chen H. Risk factors for disordered eating during early and middle adolescence: Prospective evidence from mainland Chinese boys and girls. J Abnorm Psychol. 2011;120(2):454.
- Jackson T, Chen H. Predicting changes in eating disorder symptoms among Chinese adolescents: A 9-month prospective study. J Psychosom Res. 2008; 64(1):87–95.
- Zhang J, Conwell Y, Zhou L, Jiang C. Culture, risk factors and suicide in rural China: a psychological autopsy case control study. Acta Psychiatr Scand. 2004;110(6):430–7.
- Jackson T, Chen H. Identifying the eating disorder symptomatic in China: The role of sociocultural factors and culturally defined appearance concerns. J Psychosom Res. 2007;62(2):241–9.
- Morgan JF, Reid F, Lacey JH. The SCOFF questionnaire: a new screening tool for eating disorders. West J Med. 2000;172(3):164.
- 23. Hill LS, Reid F, Morgan JF, Lacey JH. SCOFF, the development of an eating disorder screening questionnaire. Int J Eat Disord. 2010;43(4):344–51.
- Leung SF, Lee KL, Lee SM, Leung SC, Hung WS, Lee WL, et al. Psychometric properties of the SCOFF questionnaire (Chinese version) for screening eating disorders in Hong Kong secondary school students: A cross-sectional study. Int J Nurs Stud. 2009;46(2):239–47.
- Brener ND, McManus T, Galuska DA, Lowry R, Wechsler H. Reliability and validity of self-reported height and weight among high school students. J Adolesc Health. 2003;32(4):281–7.
- Ma GS, Li YP, Hu XQ, Cui ZH, Yang XG, Chen CM. Report on childhood obesity in China (2). Verification of BMI classification reference for overweight and obesity in Chinese children and adolescents. Biomed Environ Sci. 2006;19(1):1–7.
- Shepeliak N. Gender and cultural differences in body dissatisfaction and self-esteem [dissertation on the Internet]. [New York(America)]: Pace University; 2006 [cited 2006 Aug 3]. Available from: http://digitalcommons. pace.edu/honorscollege_theses/35.
- Watson D, Clark LA, Tellegen A. Development and validation of brief measures of positive and negative affect: the PANAS scales. J Pers Soc Psychol. 1988;54(6):1063.
- Stice E, Bearman SK. Body-image and eating disturbances prospectively predict increases in depressive symptoms in adolescent girls: a growth curve analysis. Dev Psychol. 2001;37(5):597.
- China NBoSo. Average salary by type of work in China 2014 [updated 2014]. Available from: http://data.stats.gov.cn/easyquery.htm?cn=C01.
- 31. Fu D, Wang J, Chen W, Bi Y. Disordered eating attitudes and behaviors and related mood states among female university students in Beijing. Chin Ment Health J. 2005;19(8):525-8.
- 32. Qian MLX. Dieting and eating disorder of female students in colleges in Beijing. Chin Ment Health J. 2002;16(11):5.
- 33. Fan Y, Duan Y, Hao L, Li Y, Yang Y. Status of eating disorders of high school students in seven cities of China. Chin J Health Educ. 2009;125:657–9.
- Luck AJ, JF ML, Reid F, O'brien A, Brunton J, Price C, et al. The SCOFF questionnaire and clinical interview for eating disorders in general practice: comparative study. BMJ. 2002;325(7367):755–6.
- 35. Carlat DJ, Camargo CA. Review of bulimia nervosa in males. Am J Psychiatry. 1991;148(7):831–43.
- Forbush K, Heatherton TF, Keel PK. Relationships between perfectionism and specific disordered eating behaviors. Int J Eat Disord. 2007;40(1):37–41.
- Jacobi C, Hayward C, de Zwaan M, Kraemer HC, Agras WS. Coming to terms with risk factors for eating disorders: application of risk terminology and suggestions for a general taxonomy. Psychol Bull. 2004;130(1):19.
- Eichen DM, Conner BT, Daly BP, Fauber RL. Weight perception, substance use, and disordered eating behaviors: Comparing normal weight and overweight high-school students. J Youth Adolesc. 2012;41(1):1–13.

- Xie B, Chou C-P, Spruijt-Metz D, Reynolds K, Clark F, Palmer PH, et al. Weight perception and weight-related sociocultural and behavioral factors in Chinese adolescents. Prev Med. 2006;42(3):229–34.
- Flett GL, Hewitt PL. Perfectionism and maladjustment: An overview of theoretical, definitional, and treatment issues. Perfectionism: Theory, research, and treatment. Washington, DC: American Psychological Association; 2002. p. 5–31.
- Stice E, Hayward C, Cameron R, Killen J. Body image and eating related factors predict onset of depression in female adolescents. J Abnorm Psychol. 2000;109:438.
- Racine SE, Keel PK, Burt SA, Sisk CL, Neale M, Boker S, et al. Exploring the relationship between negative urgency and dysregulated eating: Etiologic associations and the role of negative affect. J Abnorm Psychol. 2013;122(2):433.
- Heatherton TF, Baumeister RF. Binge eating as escape from self-awareness. Psychol Bull. 1991;110(1):86.
- Spoor ST, Bekker MH, Van Strien T, van Heck GL. Relations between negative affect, coping, and emotional eating. Appetite. 2007;48(3):368–76.
- 45. Harrison K, Cantor J. The relationship between media consumption and eating disorders. J Commun. 1997;47(1):40–67.
- Becker AE, Burwell RA, Herzog DB, Hamburg P, Gilman SE. Eating behaviours and attitudes following prolonged exposure to television among ethnic Fijian adolescent girls. Br J Psychiatry. 2002;180(6):509–14.
- Khan AN, Khalid S, Khan HI, Jabeen M. Impact of today's media on university student's body image in Pakistan: a conservative, developing country's perspective. BMC Public Health. 2011;11(1):379.
- Yamamiya Y, Shroff H, Thompson JK. The tripartite influence model of body image and eating disturbance: A replication with a Japanese sample. Int J Eat Disord. 2008;41(1):88–91.
- RYU HR, Lyle RM, MCCABE GP. Factors associated with weight concerns and unhealthy eating patterns among young Korean females. Eat Disord. 2003; 11(2):129–41.
- Becker AE. Television, disordered eating, and young women in Fiji: Negotiating body image and identity during rapid social change. Cult Med Psychiatry. 2004;28(4):533–59.
- Garfinkel PE, Garner D. M. Anorexia nervosa: A multidimensional perspective. New York: Brunner/Mazel: 1982.
- 52. Darby A, Hay P, Mond J, Rodgers B, Owen C. Disordered eating behaviours and cognitions in young women with obesity: relationship with psychological status. Int J Obes. 2007;31(5):876–82.
- Rø Ø, Reas DL, Rosenvinge J. The impact of age and BMI on Eating Disorder Examination Questionnaire (EDE-Q) scores in a community sample. Eat Behav. 2012;13(2):158–61.
- Li H, Prevatt F. Fears and related anxieties in Chinese high school students. Sch Psychol Int. 2008;29(1):89–104.
- Abebe DS, Torgersen L, Lien L, Hafstad GS. Soest Tv. Predictors of disordered eating in adolescence and young adulthood: A population-based, longitudinal study of females and males in Norway. Int J Behav Dev. 2014;38(2):128–38.
- Stommel M, Schoenborn CA. Accuracy and usefulness of BMI measures based on self-reported weight and height: findings from the NHANES & NHIS 2001-2006. BMC Public Health. 2009;9(1):421.
- Solmi F, Hatch SL, Hotopf M, Treasure J, Micali N. Validation of the SCOFF questionnaire for eating disorders in a multiethnic general population sample. Int J Eat Disord. 2015;48(3):312–6.
- 58. Shi Y, Zhang L, Ma Y, Yi H, Liu C, Johnson N, et al. Dropping out of rural China's secondary schools: A mixed-methods analysis. China Q. 2015;224:1048–69.

Submit your next manuscript to BioMed Central and we will help you at every step:

- We accept pre-submission inquiries
- Our selector tool helps you to find the most relevant journal
- We provide round the clock customer support
- Convenient online submission
- Thorough peer review
- Inclusion in PubMed and all major indexing services
- Maximum visibility for your research

Submit your manuscript at www.biomedcentral.com/submit

